UC Irvine UC Irvine Previously Published Works

Title

Decreased Thalamic Activity Is a Correlate for Disconnectedness during Anesthesia with Propofol, Dexmedetomidine and Sevoflurane But Not S-Ketamine.

Permalink

https://escholarship.org/uc/item/5nm4h9g8

Journal

The Journal of neuroscience : the official journal of the Society for Neuroscience, 43(26)

ISSN

0270-6474

Authors

Kantonen, Oskari Laaksonen, Lauri Alkire, Michael <u>et al.</u>

Publication Date

2023-06-01

DOI

10.1523/jneurosci.2339-22.2023

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed

Behavioral/Cognitive

Decreased Thalamic Activity Is a Correlate for Disconnectedness during Anesthesia with Propofol, Dexmedetomidine and Sevoflurane But Not S-Ketamine

Oskari Kantonen,^{1,2,3} Lauri Laaksonen,^{1,2} Michael Alkire,⁴ Annalotta Scheinin,^{1,2} Jaakko Långsjö,⁵ Roosa E. Kallionpää,^{2,6} Kaike Kaisti,^{1,7} Linda Radek,¹ Jarkko Johansson,^{1,8} Timo Laitio,² Anu Maksimow,² Joonas Scheinin,² Mikko Nyman,⁹ [®] Mika Scheinin,¹⁰ Olof Solin,¹ Tero Vahlberg,¹¹ Antti Revonsuo,^{6,12} [®] Katja Valli,^{2,6,12} and [®] Harry Scheinin^{1,2,10}

¹Turku PET Centre, University of Turku and Turku University Hospital, Turku FI-20521, Finland, ²Department of Perioperative Services, Intensive Care and Pain Medicine, Turku University Hospital, University of Turku, Turku FI-20521, Finland, ³Department of Perioperative Services, Intensive Care and Pain Medicine, Satakunta Central Hospital, Pori FI-28500, Finland, ⁴Department of Anesthesiology and Perioperative Medicine, University of California, Irvine, Irvine, California 92868, ⁵Department of Intensive Care, Tampere University Hospital, Tampere FI-33521, Finland, ⁶Department of Psychology and Speech-Language Pathology, and Turku Brain and Mind Center, University of Turku, Turun yliopisto FI-20014, Finland, ⁷Department of Anesthesiology and Intensive Care, Oulu University Hospital, Oulu FI-90029, Finland, ⁸Department of Radiation Sciences, Umeå University, Umeå SE-901 87, Sweden, ⁹Department of Radiology, Turku University Hospital, Turku FI-20521, Finland, ¹⁰Institute of Biomedicine and Unit of Clinical Pharmacology, University of Turku and Turku University Hospital, Turun yliopisto FI-20014, Finland, ¹¹Institute of Clinical Medicine, Biostatistics, University of Turku and Turku University Hospital, Turun yliopisto FI-20014, Finland, ¹¹Institute of Clinical Medicine, Biostatistics, University of Turku and Turku University Hospital, Turun yliopisto FI-20014, Finland, ¹¹Department of Cognitive Neuroscience and Philosophy, School of Bioscience, University of Skövde, Skövde SE-541 28, Sweden

Establishing the neural mechanisms responsible for the altered global states of consciousness during anesthesia and dissociating these from other drug-related effects remains a challenge in consciousness research. We investigated differences in brain activity between connectedness and disconnectedness by administering various anesthetics at concentrations designed to render 50% of the subjects unresponsive. One hundred and sixty healthy male subjects were randomized to receive either propofol $(1.7 \,\mu\text{g/ml}; n = 40)$, dexmedetomidine $(1.5 \,\text{ng/ml}; n = 40)$, sevoflurane (0.9% end-tidal; n = 40), S-ketamine $(0.75 \,\mu\text{g/ml}; n = 20)$, or saline placebo (n = 20) for 60 min using target-controlled infusions or vaporizer with end-tidal monitoring. Disconnectedness was defined as unresponsiveness to verbal commands probed at 2.5-min intervals and unawareness of external events in a postanesthesia interview. High-resolution positron emission tomography (PET) was used to quantify regional cerebral metabolic rates of glucose (CMR_{glu}) utilization. Contrasting scans where the subjects were classified as connected and responsive versus disconnected and unresponsive revealed that for all anesthetics, except S-ketamine, the level of thalamic activity differed between these states. A conjunction analysis across the propofol, dexmedetomidine and sevoflurane groups confirmed the thalamus as the primary structure where reduced metabolic activity was related to disconnectedness. Widespread cortical metabolic suppression was observed when these subjects, classified as either connected or disconnected, were compared with the placebo group, suggesting that these findings may represent necessary but alone insufficient mechanisms for the change in the state of consciousness.

Key words: Anesthesia; connected; consciousness; disconnected; neuroimaging; positron emission tomography

Received Dec. 24, 2022; revised Apr. 29, 2023; accepted May 8, 2023.

Author contributions: M.A., J.L., R.E.K., K.K., A.R., K.V., and H.S. designed research; O.K., L.L., A.S., J.L., R.E.K., K.K., T.L., A.M., J.S., and K.V. performed research; M.S., O.S., and T.V. contributed unpublished reagents/analytic tools; O.K., J.J., M.N., M.S., T.V., A.R., and H.S. analyzed data; O.K. wrote the first draft of the paper; O.K., LL., M.A., A.S., J.L., R.E.K., K.K., L.R., J.J., T.L., A.M., J.S., M.S., O.S., T.V., A.R., K.V., and H.S. edited the paper; O.K. and M.A. wrote the paper.

This work was supported by Academy of Finland, Helsinki, Finland Grant Numbers 266467 and 266434; Jane and Aatos Erkko Foundation, Helsinki, Finland; VSSHP-EVO Grant Numbers 13223 and L3824, Turku, Finland; Doctoral Programme of Clinical Investigation, University of Turku Graduate School, Turku, Finland (0.K., A.S., L.R.); Paulo Foundation, Espoo, Finland (A.S.); Finnish Medical Foundation, Helsinki, Finland (0.K., A.S.); The Orion Research Foundation, Espoo, Finland (A.S.); Signe and Ane Gyllenberg Foundation (0.K.); and Emil Aaltonen Foundation (0.K., R.E.K.). We thank the radiographers and anesthesia nurses at Turku PET Centre for excellent technical assistance and Ms. Saija Sirén, Lic. Phil., for the analysis of drug concentrations in plasma.

M.S. declares consultancy and contract research relationships with Orion Pharma, the original manufacturer of dexmedetomidine. Orion Pharma was not involved in the planning or execution of the current study. M.S. was also listed as inventor in Orion Pharma's United States Patent Application US5344840A from 1993, "4-substituted imidazole derivatives useful in perioperative care," now expired. M.S. declares no current competing financial interests in the study. All other authors declare no competing financial interests.

Correspondence should be addressed to Harry Scheinin at harry.scheinin@utu.fi.

https://doi.org/10.1523/JNEUROSCI.2339-22.2023 Copyright © 2023 Kantonen et al.

This is an open-access article distributed under the terms of the Creative Commons Attribution 4.0 International license, which permits unrestricted use, distribution and reproduction in any medium provided that the original work is properly attributed.

Significance Statement

Experimental anesthesia is commonly used in the search for measures of brain function which could distinguish between global states of consciousness. However, most previous studies have not been designed to separate effects related to consciousness from other effects related to drug exposure. We employed a novel study design to disentangle these effects by exposing subjects to predefined EC_{50} doses of four commonly used anesthetics or saline placebo. We demonstrate that state-related effects are remarkably limited compared with the widespread cortical effects related to drug exposure. In particular, decreased thalamic activity was associated with disconnectedness with all used anesthetics except for S-ketamine.

Introduction

Anesthetic drugs cause reversible changes in the state of consciousness and in responsiveness to external stimuli, but the underlying neural mechanisms remain incompletely understood (Scheinin et al., 2018a). Changes in multiple domains of brain function between responsive baseline and anesthetic-induced unresponsive conditions have been studied extensively and are relatively well characterized for most, commonly used anesthetics (Bonhomme et al., 2019). Yet, a central limitation in most studies exploring the neural mechanisms by which anesthetics induce changes in the global state of consciousness is the difficulty to point out which of the observed changes in neural function contribute to the altered state of consciousness (i.e., state-related effects), and which may reflect other effects caused by the administered anesthetics (i.e., nonspecified drug-effects; Scheinin et al., 2018b). However, a growing body of evidence from our group (Långsjö et al., 2012; Scheinin et al., 2018b, 2021; Kallionpää et al., 2020) and others (Xie et al., 2011; Pal et al., 2020; Casey et al., 2022) indicates that it is possible to dissociate these effects by careful study design choices, and suggests that more limited changes in neural function associate with the state-effect, than what can be observed if the nonspecified drug-effects are not controlled for.

In our recent study (Scheinin et al., 2021), we used forced awakenings during steady-state propofol and dexmedetomidine anesthesia to dissociate state-related effects from nonspecified drug-effects in a within-subject design. Positron emission tomography (PET) imaging with $[^{15}O]H_2O$ tracer revealed that the transitions between disconnected and connected states of consciousness were consistently associated with concomitant reciprocal changes in brain activity primarily in the thalamus, anterior and posterior cingulate cortices and angular gyri. Notably, the pattern separating these states was markedly different from the widespread pattern of suppressed cortical activity commonly associated with anesthetic-induced unresponsiveness (Kaisti et al., 2002; Laitio et al., 2009; Akeju et al., 2014).

While the serial awakening paradigms have previously been implemented to reveal neural mechanisms related to specific states of consciousness, the studies have been limited to using propofol and dexmedetomidine as anesthetics. The threshold-level dosing required in such experiments has proven problematic with other commonly used anesthetics, such as ketamine and sevoflurane, because of excessive cognitive impairment (Casey et al., 2022) or movement in the scanner (Palanca et al., 2015). Hence, to study the state-related effects of a wide range of anesthetics with different primary mechanisms of action, we randomized healthy male volunteers to receive EC_{50} for unresponsiveness to verbal command of either propofol, dexmedetomidine, sevoflurane or S-ketamine using a parallel-group design and applied PET imaging with ¹⁸F-fluorodeoxyglucose (FDG)

tracer to measure brain activity. Trapping of the tracer inside active brain cells creates a long-lasting imprint of the distribution of metabolic activity in the brain during the drug intervention, enabling brain scans to be acquired later when the subjects are awake and co-operative, rather than during the anesthesia (Alkire et al., 1999). A placebo group was included to allow comparisons between the drug-free wakeful state and anesthesia.

The study was designed to answer three main questions: (1) what are the neural mechanisms of disconnectedness, assessed by contrasting scans from subjects in different states of consciousness but with similar drug exposure; (2) are these state-related mechanisms shared across different anesthetics when the drug-effect is accounted for; and (3) what are the nonspecified drug-effects and combined effects of the drug and the state? Connectedness is here defined as the ability to consciously perceive external stimuli and to either perform meaningful voluntary behavioral responses or later report having experienced external events. Disconnectedness may indicate total absence of internal conscious experiences, but is also consistent with the presence of internally generated, externally disconnected conscious experiences (e.g., dreams).

Materials and Methods

Subjects

One hundred and sixty 18- to 30-year-old healthy, ASA 1 (according to the American Society of Anesthesiologists physical status classification system), right-handed volunteers participated in the study. The trial (NCT0262440) was approved by the Ethics Committee of the Hospital District of Southwest Finland and the Finnish Medicines Agency Fimea. All subjects provided written informed consent in accordance with the Declaration of Helsinki. Radiation exposure concerns related to the PET imaging limited the subjects to only male volunteers. All subjects were interviewed and examined by a licensed physician (L.L., A.S., or O.K.). A standard 12-lead electrocardiogram (ECG) and blood and urine samples were also taken. Exclusion criteria included any somatic illness, chronic medication or drug allergy, history of any psychiatric disorder or substance abuse, cardiac arrhythmias, hearing impairment, serious nausea in connection with previous anesthesia, blood donation within 90 d before the study, prior participation in a PET/SPECT study, any contraindication to magnetic resonance imaging (MRI), and pathologic findings in laboratory tests or positive urine drug screening.

Experimental design

We used the PET neuroimaging data, responsiveness test data and structured interview data from our recently published studies (Laaksonen et al., 2018; Radek et al., 2021) to investigate differences in cerebral metabolic rates of glucose (CMR_{glu}), as a surrogate measure for neuronal activity (Varrone et al., 2009), between connectedness and disconnectedness, and between exposure to anesthetic agents and placebo. The design of the study is schematically illustrated in Figure 1.

Based on our previous studies, we estimated a target concentration which would render 50% of the subjects unresponsive to verbal command, i.e., EC_{50} , for each study drug. The selected EC_{50} target values were: 1.7 µg ml⁻¹ for propofol, 1.5 ng ml⁻¹ for dexmedetomidine, 0.9%



Figure 1. Design of the experiment. Each subject within each treatment group received a targeted EC_{50} concentration of either propofol, dexmedetomidine, sevoflurane, or S-ketamine. After a 20-min stabilization phase, FDG tracer was injected followed by a 40-min steady-state anesthesia phase, allowing brain FDG uptake to be completed. Then, administration of the study treatment was terminated, and an attempt was made to rouse the unresponsive subjects, followed by a brief interview to probe for mental content during the anesthesia. Next, the subjects were moved to the PET scanner for brain imaging. Responsiveness was tested every 2.5 min during the steady-state anesthesia. Behavioral states of interest: W = wakeful baseline, UR = unresponsive, R = responsive. See the Materials and Methods section for details.

end-tidal for sevoflurane and $0.75 \,\mu g \text{ ml}^{-1}$ for S-ketamine (Långsjö et al., 2005, 2012; Kaskinoro et al., 2011; Scheinin et al., 2021). The subjects received individually programmed, targeted EC50 infusions (or end-tidal %) of either propofol (n = 40), dexmedetomidine (n = 40), sevoflurane (n = 40), S-ketamine (n = 20), or saline placebo (n = 20). The participants were randomly allocated to the different treatments, but the treatments were not masked. Permuted blocks were applied to achieve balanced groups across treatments. The anesthetic protocol was initiated with a 20-min stabilization phase, after which the FDG PET radiotracer was injected, and the achieved pseudo-steady-state phase was maintained for 40 min. Responsiveness to verbal command was tested every 2.5 min during anesthesia and arterial blood samples were drawn for drug concentration measurements and tracer kinetic modeling. Brain retention of the FDG tracer was considered to be stabilized after the 40 min pseudosteady-state anesthesia phase (Alkire et al., 1997). During this period, the FDG tracer was taken up to metabolically active brain cells and converted to [¹⁸F]FDG-6-phosphate via hexokinase-mediated phosphorylation (Duelli and Kuschinsky, 2001; Varrone et al., 2009). Because [¹⁸F] FDG-6-phosphate cannot be metabolized further and has low membrane permeability, it becomes trapped inside the cells with a very slow clearance (Gallagher et al., 1978; Phelps et al., 1979). Because of these tracer properties, the subsequent PET scan effectively reflects glucose metabolism in the brain during the preceding tracer uptake period (Alkire et al., 1999). After terminating the administration of the treatment, subjects were awakened (if unresponsive), and a detailed interview of subjective experiences was conducted for all subjects. Finally, a PET scan measuring absolute CMRglu was acquired. One scan and one interview were collected for each subject.

Anesthesia protocol

The subjects abstained from the use of alcohol and any medications for at least 48 h and from the use of caffeine-containing products for 10– 12 h and fasted overnight before the experiments. Two forearm veins were cannulated for administration of study treatments and the PET radiotracer. The left radial artery was cannulated to obtain blood samples for drug concentration measurements and kinetic modeling of the PET radiotracer. The intravenous anesthetic agents, propofol (Propofol Lipuro 10 mg/ml, B. Braun), dexmedetomidine (Dexdor 100 μ g/ml, Orion Pharma) and S-ketamine (Ketanest-S 25 mg/ml, Pfizer) were administered using target controlled infusion (TCI) with previously described pharmacokinetic parameters (Domino et al., 1984; Marsh et al., 1991; Talke et al., 2003). A Harvard 22 syringe pump (Harvard Apparatus) and a portable computer running Stanpump software (by Steven L. Schafer, MD; www.opentci.org/code/stanpump) were used for drug administration. Plasma target levels were used for the intravenous anesthetics. Sevoflurane (Sevorane 100%, AbbVie) was administered through a tight-fitting face mask using a standard anesthesia work-station (Kaskinoro et al., 2011) with fresh gas flow set at 6 l min⁻¹.

Responsiveness test and classification of the state of consciousness

During the experiment, the subjects were lying on bed with custommade response handles secured to their wrists. Responsiveness was tested with a standardized prerecorded auditory request "press the handles twice" (R-test). Responding was practiced before the anesthesia. The R-test was presented with Presentation 17.0 stimulus delivery and experimental control software system (Neurobehavioral Systems Inc), and stimuli were delivered via headphones. The request was presented once at 2.5-min intervals throughout the experiment. A response to the R-test categorized the subject as responsive for the subsequent 2.5-min period. No response to R-test resulted in the subject being classified as unresponsive for the subsequent 2.5-min period.

Targeting an EC₅₀ dose might imply that 50% of subjects will be unresponsive during the entire PET tracer uptake time and that 50% would remain responsive. Alternatively, attempting to hold subjects on the borderline of being conscious of external events, one might expect fluctuations between states of responsiveness could occur one or more times during the tracer uptake period (McKinstry-Wu et al., 2019; Wasilczuk et al., 2020). Hence, the categorization of each subject as responsive or unresponsive may depend in part on how much time each person spent in the predominant state of responsiveness. Furthermore, since the overall rate of tracer uptake in active brain cells varies as a function of tracer concentration in plasma, the time period right after the tracer injection (when the tracer concentration in the plasma is high) will influence the subsequent PET image more than the time periods toward the end of the experiment (when the tracer concentration in plasma is low). To address this, we modeled the relative FDG brain uptake for each R-test period using blood samples from the placebo group drawn during the experiment for tracer kinetic modeling. Based on this model, a weight was assigned to each R-test period reflecting their proportional significance in forming the final CMR_{glu} image used for statistical testing. For each subject, the values of individual R-test weights were then summed for both responsive and unresponsive trials to form estimations of total FDG uptake, and subsequently PET imaging signal in each respective state of responsiveness. Greater than $0.8 \times$ total

FDG brain uptake while unresponsive (UR) during the 40-min steadystate drug exposure was chosen as a threshold to assign a subject to this behavioral category. Greater than $0.4 \times$ total FDG uptake in the responsive (R) state was chosen as a threshold to assign a subject to this category. In case a subject would not qualify for either category (i.e., the state was too labile for the subject to be categorized as responsive or unresponsive) they would be assigned to a third category, unclassified (UC). Note that the classification of the subjects into responsiveness classes differs here from Radek et al. (2021), where the subjects were classified as responsive or unresponsive based on the last two R-test results for optimization of interview analyses. Here, the classification procedure was optimized for PET analysis so that it would best reflect the amount of tracer uptake (and thus PET imaging signal) in each responsiveness category.

The categorization of subjects as connected or disconnected was based on (1) responsiveness classification and (2) structured interviews conducted at the end of the treatment period (Radek et al., 2021). Subjects classified as responsive were considered connected and conscious. Subjects classified as unresponsive with no explicit report of external perceptual content in the structured interview were considered disconnected.

Magnetic resonance imaging (MRI)

For each subject, an anatomic MRI scan of the brain (T1, T2, FLAIR) was performed before the PET imaging session for subsequent image preprocessing and exclusion of any brain anomalies. MRI scans were obtained with a Philips Ingenuity PET-MR 3T scanner (Philips Medical Systems) with the following scan parameters: TR = 25 ms, TE = 4.6 ms, flip angle 30°, scan time 376 s. A trained neuroradiologist (MN) evaluated the anatomic images for any pathologic findings.

Positron emission tomography (PET) imaging

PET imaging was performed using an ECAT HRRT scanner (Siemens CTI; Eriksson et al., 2002). The HRRT is a dual-layer, LSO-LYSO crystal-detector scanner characterized by a nearly isotropic 2.5-mm intrinsic spatial resolution. In the reconstructed images spatial resolution is 2.5–3 mm in the radial and tangential directions and 2.5–3.5 mm in the axial direction in the 10 cm field-of-view covering most of the brain (de Jong et al., 2007). Subjects were positioned supine in the scanner, wearing an individual thermoplastic mask to minimize head movements. Head motion was monitored with a high-precision, stereotaxic tracking device (Polaris Vicra, Northern Digital).

¹⁸F was produced by irradiating enriched [¹⁸O]H₂O with protons (cyclotron CC 18/9, D.V. Efremov Institute, St. Petersburg, Russia) and an automated device (Fastlab; GE Healthcare) was used in [¹⁸F] FDG synthesis according to Good Manufacturing Practice regulations. Radiochemical purity of the product exceeded 95%. A 300 MBq dose of [¹⁸F]FDG was administered over 15 s by an automated infusion system (Rad Injector, Tema Sinergie). After the tracer uptake period, administration of the study treatment was terminated, and a 30-min PET scan was performed followed by a CT transmission scan. PET data acquisition was initiated \sim 45–50 min after tracer injection in list mode format. Images were formed using an iterative OP-OSEM algorithm with resolution modeling (12 iterations, 16 subsets, including corrections for attenuation, scatter, and random events). PET data were histogrammed into a single 30-min frame reflecting the assumption of stable [18F]FDG retention over this late time window. In case the measured head movement exceeded 2.5 mm, scans were divided into subframes and subsequently co-registered into a single 30 min frame by using a multiple acquisition frame image reconstruction procedure and an attenuation correction realignment algorithm (Johansson et al., 2016). This procedure was used in four dexmedetomidine scans, two S-ketamine scans, and two sevoflurane scans.

Drug concentration measurements

Arterial blood samples for drug concentration analysis in plasma were drawn at baseline and at 20, 30, 40, and 50 min after the start of the administration of the intravenous anesthetics. High-performance liquid chromatography with tandem mass spectrometry (HPLC-MS/MS) was

used for dexmedetomidine and S-ketamine. Propofol concentrations were measured with HPLC and fluorescence detection. Interassay coefficients of variation in the relevant concentration ranges were 1.2–2.9%, 3.7–8.0%, and 0.7–2.2%, respectively (Laaksonen et al., 2018). Online end-tidal anesthetic gas concentration monitoring was used for sevoflurane.

PET data preprocessing

Voxel-wise maps of CMR_{glu} were generated on the basis of average [¹⁸F] FDG retention at 50–80 min after tracer injection and plasma radioactivity data. Individual MRIs were co-registered with the subjects' parametric brain images using SPM8. Voxel-wise maps of CMR_{glu} were calculated according to the following equation:

$$CMR_{glu} = \frac{FUR \times C_{glu}}{LC}$$

where FUR is the fractional uptake rate of [¹⁸F]FDG relative to the integral (0–90 min) of the [¹⁸F]FDG concentration in plasma, C_{glu} is the plasma glucose concentration, and LC is the lumped constant (0.65; Wu et al., 2003). Assuming an average brain tissue density of 1.04 g ml⁻¹, CMR_{glu} values were converted into units of µmol 100 g⁻¹ min⁻¹.

Nonlinear mapping from the MRI to the MNI standard space was estimated using unified segmentation in SPM8, and the deformations were subsequently applied to the MRI and co-registered PET images. All normalized PET images were smoothed using an isotropic Gaussian kernel of 12-mm FWHM.

Statistical analysis

The Kolmogorov-Smirnov test was applied to test the normality of the outcome variables. Pearson's χ^2 test (χ^2) was used to compare responsiveness between the treatments followed by *post hoc* χ^2 tests between treatment pairs. The Kruskal-Wallis H test (H) with the Dwass, Steel, Critchlow-Flinger method in pairwise comparisons was used to compare simulation-based FDG uptake weights for subjects assigned to the responsive or unresponsive categories between the treatments. The mean measured drug concentrations between the responsive and unresponsive subjects was compared with two-sample t test. Voxel-averaged regional CMR_{glu} values were extracted within a 2-mm sphere region of interest (ROI) centered at the peak effect location from the SPM conjunction analysis (see below) using the MarsBar toolbox (Brett et al., 2002). Two-way ANOVA was used to analyze differences in absolute ROI CMR_{ghu} values between the connected and disconnected subjects followed by contrasts within the drugs. Pearson correlation coefficients (*r*) and coefficients of determination (R^2) were calculated to assess the associations of the modeling-based weights for UR and mean measured drug concentrations with the $\mathrm{CMR}_{\mathrm{glu}}$ from the previously described ROI. When applicable, Bonferroni correction was used to correct for multiple comparisons. A two-tailed p < 0.05 was considered statistically significant. Results are given as means (SD) if not otherwise stated. Statistical analyses, except for the voxel-wise neuroimaging data analyses (see below), were performed with SAS System for Windows, version 9.4. (SAS Institute Inc.).

A flexible factorial model was constructed in SPM12 with treatment group and state (connected or disconnected) as factors. Two-way interactions between factors were modeled in the design matrix, and t-contrasts were generated to assess differences in regional brain activity between subjects classified as connected versus disconnected (i.e., staterelated effects) within each treatment group. Conjunction analysis was used to investigate common and jointly significant state-related effects across selected treatments (Friston et al., 2005). Combined effects of drug and state were investigated by contrasting placebo with the disconnected state, and drug-effects only, by contrasting placebo with the connected state. Because of the ad hoc nature of the thresholds of simulated FDG uptake weight by which the subjects were assigned to behavioral categories R, UR, or UC, a supplementary continuous analysis using a multiple regression model with simulated uptake weights for the UR condition as covariates was conducted. Thus, as the UR weight reflects the proportion of total FDG uptake in the disconnected state, the regression



Figure 2. Responsiveness during the experiment. *A*, Results of responsiveness tests performed every 2.5 min during the 40-min tracer uptake period and estimated EC_{50} steady state administration of each study drug. *B*, Tracer kinetic modeling-based weights for each 2.5-min R-test period are displayed above R-test results from one representative subject for each responsiveness category. The weights were combined with R-test results to form an estimation of the proportion of total tracer uptake, and subsequently PET image signal, in either responsive (R_w) or unresponsive (UR_w) state as the sum of weights in each respective state (for details, see Extended Data Fig. 2-1). *C*, *D*, Boxplots of cumulative R-test weights for subjects classified as responsive (blue boxplots) or unresponsive (red boxplots) in each active treatment group. The median R weight of subjects classified as responsive was lower in the dexmedetomidine group than in the other treatment groups, but only reached statistical significance in pairwise comparison with the propofol group (*p = 0.0155). The median UR weight of subjects classified as unresponsive was significantly lower in the S-ketamine group compared with the sevoflurane group (°p = 0.0153). Boxes represent lower quartiles, medians and upper quartiles, and whiskers represent 1.5× interquartile ranges above and below the upper and lower quartiles. N/A, missing responsiveness data.

analysis tests for significant associations between disconnectedness and regional CMR_{glu}. Global normalization with proportional scaling was applied to account for interindividual variability in CMR_{glu} and global effects of anesthetics (Alkire et al., 2000). Clusters of voxels at p < 0.05 FWE corrected were considered as significant, starting with an initial voxel-level threshold of p < 0.001. Results are presented at a significance threshold of p < 0.001, uncorrected, in case the findings did not survive the correction for multiple comparisons, but were localized within regions previously associated with consciousness-related metabolic effects (Alkire et al., 2000; Långsjö et al., 2005). Because of increased statistical power in the conjunction analyses, a more stringent threshold of p < 0.05 FWE corrected at voxel level was used to allow localization of peak effects. Conjunction analyses were tested against the global null hypothesis. MRIcroGL (https://www.nitrc.org/projects/ mricrogl/) and Surfice (https://www.nitrc.org/projects/surfice) were used for visualization. The Human Brainnetome Atlas (Fan et al., 2016) and the Atlas of the Human Brain (Mai et al., 2016) were used for localization of the findings in the MNI space.

Results

Behavioral state, subjective reports, and ensuing classification of the state of consciousness

The subjects were prone to fluctuation in responsiveness because of the administered EC_{50} anesthetic dose, and they tended to become more unresponsive toward the end of the drug exposure phase (Fig. 2A). Weights based on tracer uptake kinetic modeling (Fig. 2B) were applied to assign subjects to responsiveness classes (R, UR, or UC; see Materials and Methods and Extended Data Fig. 2-1 for details). The final number of achieved states and cumulative model-based uptake weights for each active treatment group are presented in Table 1. The proportions of responsive, unresponsive, and unclassified subjects were significantly different in the sevoflurane group compared with the propofol and S-ketamine groups (p = 0.0024 and p = 0.016; Table 1). The median uptake weights for subjects classified as responsive were significantly different between the propofol and dexmedetomidine groups (p = 0.0115; Table 1; Fig. 2C). The median uptake weights for subjects classified as unresponsive were significantly different between the sevoflurane and S-ketamine groups (p = 0.0153; Table 1; Fig. 2D). Extended Data Figure 2-1 summarizes the R-test results in each study subject during the experiment, the calculated FDG brain uptake weight and the ensuing classification.

After drug administration, 11 (out of 15), 18 (out of 20), and 11 (out of 30) unresponsive subjects from the propofol, dexmedetomidine and sevoflurane groups, respectively, were successfully interviewed for mental content during the preceding unresponsive period. Disconnected experiences, i.e., experiences taking place during unresponsiveness but not directly associated with external perceptual content, such as dreams, were reported by 4/11 (36%), 15/18 (83%), and 5/11 (45%) of subjects in the propofol, dexmedetomidine and sevoflurane groups, respectively. None of the unresponsive subjects in the S-ketamine group could be interviewed immediately after terminating the

Tab	le	1.	Responsivene	ss state	categorization	and	cumulative	tracer	uptake	weights
-----	----	----	--------------	----------	----------------	-----	------------	--------	--------	---------

	Responsive				Unclassified	
Drug	n (%)	Uptake weight in responsive state	n (%)	Uptake weight in unresponsive state	n (%)	Uptake weight in responsive state
Propofol $n = 40$	21 (52.5%)	0.84 (0.58-0.94)	15 (37.5%)	1.00 (0.95–1.00)	4 (10%)	0.31 (0.27–0.33)
Dexmedetomidine $n = 40$	13 (32.5%)	0.51 (0.49–0.57)	20 (50%)	0.98 (0.90-1.00)	7 (17.5%)	0.37 (0.33-0.38)
Sevoflurane $n = 40$	5 (12.5%)	0.69 (0.67-0.88)	30 (75%)	1.00 (0.97-1.00)	5 (12.5%)	0.13 (0.07-0.19)
S-ketamine <i>n</i> = 20	10 (50%)	0.83 (0.61-0.94)	7 (35%)	0.94 (0.85–0.96)	3 (15%)	0.33 (0.25-0.34)

The final number (proportions) of subjects assigned to responsiveness state categories and median (IQR) uptake weights for respective categories in each treatment group. The overall proportions of responsive, unresponsive, and unclassified subjects were significantly different between treatment groups ($\chi^2(6) = 18.4$, p = 0.0052). In pairwise comparisons, the corresponding proportions were significantly different in the sevolfurane group compared with the propofol and S-ketamine groups ($\chi^2(2) = 15.0$, p = 0.0024) and ($\chi^2(2) = 11.0$, p = 0.016), respectively. There was a significant difference in median uptake weights between treatment groups in responsive and unresponsive subjects (H(3) = 11.9, p = 0.0024) and ($\chi^2(2) = 11.0$, p = 0.015), respectively. In pairwise comparisons, the difference was significant between propoli and dexmedetomidine groups in responsive subjects (H(3) = 11.9, p = 0.0076) and (H(3) = 9.8, p = 0.0204), respectively. In pairwise comparisons, the difference was significant between propoli and dexmedetomidine groups in responsive subjects (Z = 3.1, p = 0.0115, corrected with the Dwass, Steel, Critchlow-Flinger method) and between sevoflurane and S-ketamine groups in unresponsive subjects (Z = 3.0, p = 0.0153, corrected with the Dwass, Steel, Critchlow-Flinger method).

Table 2. Targeted and measured drug concentrations during the experiment

	Targeted	Responsive	Unresponsive	Two-sample
Drug	concentration	Measured	Measured	t test
Propofol (µg/ml)	1.7 (0.0)	1.65 (0.35)	1.89 (0.33)	$t_{(34)} = -2.06 \ p = 0.190$
Dexmedetomidine (ng/ml)	1.5 (0.0)	2.09 (0.40)	2.16 (0.38)	$t_{(31)} = -0.53 \ p = 1.00$
Sevoflurane (% end-tidal)	0.90 (0.0)	0.90 (0.01)	0.91 (0.03)	$t_{(33)} = -1.22 \ p = 0.919$
S-ketamine (μ g/ml)	0.75 (0.0)	0.93 (0.15)	1.05 (0.20)	$t_{(15)} = -1.41 \ p = 0.720$

Mean (SD) targeted and measured drug concentrations in plasma or end-tidal gas in the propofol, dexmedetomidine, sevoflurane and S-ketamine groups. No statistically significant differences were observed in the measured concentrations between the responsive and unresponsive subjects (two-tailed p > 0.05, Bonferroni corrected).



B Conjunction: Propofol, Dexmedetomidine and Sevoflurane



Figure 3. State-related effects. *A*, Relative CMR_{glu} differences between connectedness and disconnectedness within each treatment group representing state-related effects for each drug. Cool colors represent reduced, and warm colors increased relative glucose metabolism in the disconnected state as contrasted to the connected state. In the propofol and sevoflurane groups, decreased thalamic and medial temporal cortex activity was associated with disconnectedness (p < 0.05, FWE corrected at cluster level). In the dexmedetomidine group, a state-related effect of decreased activity was found in the left midline thalamus (p < 0.001, uncorrected). In the S-ketamine group, disconnectedness was associated with increased activity at the intersection of the left intraparietal sulcus and dorsal angular gyrus (p < 0.001, uncorrected). **B**, A conjunction analysis of the contrasts between connectedness and disconnectedness across propofol, dexmedetomidine and sevoflurane groups revealed a significant overlapping effect of decreased CMR_{glu} in the absence of connected consciousness bilaterally in the thalamus, with largest reductions in the left central medial thalamic nucleus (p < 0.05, FWE corrected at voxel level). The colorbar depicts *t* values. Similar results were obtained in a regression analysis using each subject's UR weight (a measure of disconnectedness) as a regressor and brain metabolism as a response variable (Extended Data Fig. 3-1).

drug infusion because of residual drug-effects. There were no reports indicating connectedness. Thus, all unresponsive subjects were classified as disconnected, whereas all responsive subjects were classified as connected and conscious.

Drug exposure

There were no statistically significant differences between measured drug concentrations in plasma (or end-tidal air) between the connected and disconnected subjects within the treatment groups (p > 0.05; Table 2). Thus, any differences in brain metabolism between connected and disconnected subjects may be considered to be related to the difference in the compared states.

State-related effects

In the propofol group, disconnectedness was associated with relatively lower glucose metabolism bilaterally in the thalamus, hippocampus, amygdala, nucleus accumbens, nucleus caudatus, and

> globus pallidus (p < 0.05, FWE corrected at cluster level; Fig. 3A). Within the sevoflurane group, similar state-related CMR_{glu} differences were found bilaterally in the thalamus and the cerebellum, and in the right hippocampus, amygdala, anterior parahippocampal gyrus, temporal fusiform cortex and globus pallidus. In the dexmedetomidine and S-ketamine groups, no statistically significant differences were found after correcting for multiple comparisons. However, with a significance threshold of p < 0.001, uncorrected, disconnectedness was associated with lower activity in the left midline thalamus in the dexmedetomidine group, and higher activity at the intersection of the left intraparietal sulcus and angular gyrus in the Sketamine group.

> Next, we used conjunction analysis to investigate consistent effects related to disconnectedness across treatments. Since the within-treatment group comparisons suggested different consciousness-related mechanisms for S-ketamine compared with other treatments, only propofol, dexmedetomidine and sevoflurane groups were included. An overlapping effect of decreased metabolism was observed mainly in the thalamus bilaterally, and to lesser extent in the left anterior temporal fusiform gyrus (p < 0.05, FWE corrected at voxel level; Fig. 3B). The peak effect was located at the left central medial thalamic nucleus (CMT), while significant decreases were also observed in other

adjacent thalamic nuclei.

Since the thresholds for state classification were defined *ad hoc*, we performed a confirmatory regression analysis using each

subject's classification weight for unresponsiveness as an explanatory variable. These results largely converged with those observed using dichotomized states: in the propofol, dexmedetomidine and sevoflurane groups, disconnectedness was negatively correlated with CMR_{glu} mainly in the thalamus, while in the S-ketamine group there was a positive correlation between disconnectedness and CMR_{glu} in the left intraparietal sulcus and left angular gyrus (p < 0.05, FWE corrected at cluster level)for propofol and sevoflurane groups and p < 0.001, uncorrected for dexmedetomidine and S-ketamine groups; Extended Data Fig. 3-1A). Conjunction analysis across propofol, dexmedetomidine and sevoflurane groups revealed overlapping negative correlation between disconnectedness and CMR_{glu} mainly in the thalamus bilaterally (p < 0.05, FWE corrected at voxel level; Extended Data Fig. 3-1B).

Next, CMR_{glu} values were extracted within the thalamic peak effect location from the conjunction analysis (Fig. 3*B*) to explore absolute metabolic differences between connectedness and disconnectedness. In two-way ANOVA, there was a significant drug by state interaction ($F_{(3,113)} = 7.2$, p = 0.0002). In pairwise comparisons within drugs, the effect of state was significant in the propofol ($t_{(113)} = 8.3$, p < 0.0001), dexmedetomidine ($t_{(113)} = 2.97$, p = 0.0144 and sevoflurane ($t_{(113)} = 3.26$, p = 0.0056) groups, but not in the S-ketamine group ($t_{(113)} = -1.01$, p = 0.95) after Bonferroni

correction (Fig. 4). There was a significant negative correlation between the estimated proportion of tracer uptake in the disconnected state and thalamic CMR_{glu} in the propofol, dexmedetomidine and sevoflurane groups ($p \le 0.002$), but not in the S-ketamine group (Fig. 5A). The drug concentrations were also negatively correlated with thalamic CMR_{glu} in the propofol and dexmedetomidine groups ($p \le 0.034$; Fig. 5B). However, the overall differences in drug concentrations in plasma explained relatively little of the variance in CMR_{glu} in comparison with the variance explained by the state of consciousness.

Drug-related effects

To identify nonspecified drug-induced effects on brain activity, we contrasted each connected subjects' scans within each drug group with the placebo group scans. We found that exposure to propofol, dexmedetomidine or sevoflurane was associated with relative suppression of CMR_{glu} bilaterally in the posterior cingulate cortex, precuneus, inferior parietal cortex and occipital cortex (p < 0.05, FWE corrected at cluster level; Fig. 6). Propofol and dexmedetomidine induced additional suppression bilaterally in the dorsolateral prefrontal cortex. Dexmede-tomidine additionally suppressed CMR_{glu} bilaterally in the thalamus, dorsomedial prefrontal cortex and ventromedial prefrontal cortex. Exposure to S-ketamine was associated with relative decreases of CMR_{glu} bilaterally in the insular cortex, ventromedial prefrontal cortex, occipital



Figure 4. Glucose metabolism in the central medial thalamus in connected and disconnected states. Boxplots for absolute glucose metabolism at the peak effect location (central medial thalamic nucleus) from the conjunction analysis. Thalamic CMR_{glu} was significantly different between connectedness and disconnectedness in the propolol (*p < 0.0001), dexmedetomidine (°p = 0.0144) and sevoflurane groups ($\Diamond p = 0.0056$). Boxes represent lower quartiles, medians and upper quartiles, and whiskers represent $1.5 \times$ interquartile ranges above and below the upper and lower quartiles, respectively. The middle of the confidence diamonds represents the mean, and the top and the bottom represent the upper and lower 95% confidence limits of the mean.

cortex, cerebellum, inferior temporal cortex, globus pallidus, nucleus accumbens, and left temporal fusiform cortex.

Combined effects of drug and state

Contrasting each disconnected subjects' scans within each drug group with the placebo group scans, we found that propofol, dexmedetomidine and sevoflurane induced relative suppression of CMR_{glu} bilaterally in the thalamus, posterior cingulate cortex, precuneus, inferior parietal cortex, occipital cortex and dorsolateral prefrontal cortex (p < 0.05, FWE corrected at cluster level; Fig. 7). Dexmedetomidine additionally suppressed CMR_{glu} in the dorsal and ventral medial prefrontal cortices. Compared with placebo, S-ketamine induced relative increases in CMR_{glu} in the left supramarginal gyrus, bilateral superior parietal lobule and right precuneus. Relative CMR_{glu} decreases were noted bilaterally in the lingual and primary visual cortices, and in the brainstem, more specifically in the pons at the site of the locus coeruleus, and in the posterior aspect of the medulla.

Discussion

We found that decreased metabolic activity of the thalamus was specifically associated with disconnectedness during EC_{50} administration of either propofol, dexmedetomidine or sevoflurane. Subjective experiences were frequently reported on awakening implicating that unresponsiveness denoted a state of disconnectedness but not necessarily unconsciousness. The specific effects related to the



Figure 5. Thalamic glucose metabolism was more strongly correlated with the state of consciousness than with the measured drug concentration. *A*, The estimated proportion of tracer uptake in the disconnected state (i.e., UR weight) was negatively correlated with glucose metabolism in the central thalamus in the propofol, dexmedetomidine and sevoflurane groups, but not in the S-ketamine group (r(38) = -0.790, p < 0.0001; r(38) = -0.541, p = 0.0003; r(38) = -0.491, p = 0.002 and r(18) = 0.161, p = 0.499, respectively). *B*, Correlations between mean measured drug concentration and thalamic CMR_{glu} were weaker but statistically significant in the propofol and dexmedetomidine groups (r(38) = -0.398, p = 0.012 and r(38) = -0.335, p = 0.034, respectively). r = Pearson correlation coefficient, R² = coefficient of determination, dashed lines depict the 95% confidence intervals of the regression lines.



Figure 6. Drug-related effects. Relative rCMR_{glu} differences between placebo and connected subjects from each treatment group representing nonspecified drug-effects. Cool colors represent lower relative CMR_{glu} in the treatment groups. Propofol, dexmedetomidine and sevoflurane caused relative suppression of CMR_{glu} bilaterally in the posterior cingulate cortex, precuneus, inferior parietal cortex and occipital cortex. Additionally, propofol and dexmedetomidine induced suppression of CMR_{glu} bilaterally in the dorsolateral prefrontal cortex. Dexmedetomidine additionally suppressed CMR_{glu} bilaterally in the thalamus, dorsomedial prefrontal cortex, and ventromedial prefrontal cortex. S-ketamine induced relative decrease of glucose metabolism bilaterally in the insular cortex, ventromedial prefrontal cortex, cerebellum, right inferior temporal cortex, left temporal pole, cerebellum, and brainstem areas. Statistical threshold set at p < 0.05 for all contrasts using FWE correction for multiple comparisons at cluster level, colorbar depicts *t* values.

difference in the global state of consciousness were remarkably limited compared with the combined effects of the drug and the state, or the nonspecified drug-effects alone, which were both associated with much broader changes in brain activity in the frontal, parietal, temporal, and occipital cortical areas. Our findings also suggest that S-ketamine may differ from other investigated anesthetics not only in terms of the overall or nonspecified drug-effects, but also in terms of the specific mechanisms related to changes in the state of consciousness. Several studies and theories of anesthetic mechanisms of action have indicated a critical role for the thalamus in regulating consciousness (Alkire et al., 2000; Alkire and Miller, 2005; John and Prichep, 2005; Ward, 2011; Mashour and Alkire, 2013). Early neuroimaging studies suggested that suppression of neuronal activity in the thalamus may represent a unified neural correlate for the change in the state of consciousness induced by several anesthetics with different primary mechanisms of action (Fiset et al., 1999; Alkire et al., 2000). In this view, the anesthetic



Figure 7. Combined effects of drug and state. Relative CMR_{glu} differences between placebo and disconnected subjects from each treatment group representing combined effects of drug and state. Cool colors represent lower, and warm colors higher relative CMR_{glu} in the treatment groups. Compared with the placebo group, disconnected subjects from propofol, dexmedetomidine and sevoflurane groups expressed a widespread pattern of reduced CMR_{glu} over bilateral cortical and subcortical areas. No significant increases in relative CMR_{glu} were observed for either propofol, dexmedetomidine or sevoflurane. In contrast, disconnected subjects in the S-ketamine group had relatively increased CMR_{glu} in left supramarginal gyrus and bilateral superior parietal lobule extending to right precuneal area (not visualized). Disconnected S-ketamine subjects also had lower CMR_{glu} in lingual gyrus, cerebellum and brainstem. Statistical threshold was at p < 0.05 for all contrasts using FWE correction for multiple comparisons at cluster level, colorbar depicts *t* values.

action either directly or indirectly leads to hyperpolarization of thalamocortical neurons, switching their primary firing mode from tonic to burst firing (Alkire et al., 2000; Llinás and Steriade, 2006). Much like in nonrapid eye movement sleep, this would disrupt the information transfer in the thalamo-cortico-thalamic system leading to unawareness of external perceptual events. However, in the previous studies, the data comprised both the state-related and drug-related effects introducing a possible confound when interpreting the results. The current study resolves this shortcoming and specifically supports a role for the thalamus in regulating the state of consciousness.

Transitions between global states of consciousness during constant-dose anesthesia have previously been associated with brain activity changes in the thalamus, midline cortical areas and inferolateral parietal cortex (Xie et al., 2011; Långsjö et al., 2012; Scheinin et al., 2021). Because of FDG-tracer kinetics, the results of the current study reflect averaged brain activity over the 40min steady-state drug exposure as opposed to the short period reflecting brain activity immediately after the state transition monitored with [¹⁵O]H₂O perfusion imaging. We speculate that this is the principal reason why the results of the current study primarily pinpoint effects in the thalamus, as opposed to the broader network of brain regions revealed in the previous studies. This could implicate that while the activity of a network comprising the thalamus, anterior and posterior cingulate gyri and bilateral angular gyri associates with transitions between disconnected and connected states of consciousness within a shorter time frame (Scheinin et al., 2021), suppression of thalamic neuronal activity is crucial for preventing connected consciousness.

As cortical reactivity to sensory stimulation is relatively preserved in the primary sensory areas during anesthesia (Warnaby et al., 2016; Lichtner et al., 2018; Nourski et al., 2018), the current finding of suppressed thalamic activity is better understood as a mechanism contributing to disrupted thalamo-cortico-thalamic information processing (Alkire et al., 2008) rather than a switch preventing information from reaching the cortex in the first place. This view is in agreement with both empirical findings and computational models of altered thalamocortical interactions during anesthesia and sleep (White and Alkire, 2003; Hill and Tononi, 2005; Boveroux et al., 2010; Liu et al., 2013; Vijayan et al., 2013). Supporting this view, we found the peak suppression of thalamic metabolism at the CMT and adjacent medial thalamic nuclei, which have diffuse projections to the cortex and striatum, and are associated with modulation of cortical activity patterns, arousal and cognitive functions (Saalmann, 2014; Schmitt et al., 2017). Consistently, reversal of general anesthesia has been accomplished by CMT microinfusion of either nicotine (Alkire et al., 2007) or a specific antibody blocking a type of voltage-gated potassium channels (Alkire et al., 2009). Recently, electrical stimulation of rostral intralaminar thalamic nuclei has been shown to promote arousal in anesthetized macaques while shifting cortical low-frequency and high-frequency power, spiking and coherence (Redinbaugh et al., 2020; Bastos et al., 2021), and static and dynamic functional connectivity (Tasserie et al., 2022) toward awake baseline dynamics. Furthermore, bilateral lesion of intralaminar thalamic nuclei has been associated with impairment of consciousness (Långsjö et al., 2016).

Our findings with propofol and sevoflurane suggest that preferential cortical suppression, as witnessed in sedated but responsive subjects, precedes the transition from a connected to a disconnected state. This likely reflects a combination of direct and indirect (mediated through inhibition of the arousal nuclei) cortical drug-effects. Instead, preferential suppression of the thalamus was specifically associated with disconnectedness. With dexmedetomidine, we found the thalamus to be preferentially suppressed already in responsive subjects, possibly reflecting inhibition of arousal pathways through α_2 -adrenoceptor activation-mediated disinhibition of the ventrolateral preoptic area. This view is supported by Baker et al. (2014), who reported increased δ power in CMT and cortex with dexmedetomidine before loss of righting reflex in rats. Consistently, subjects classified as connected were less responsive during the tracer uptake in the dexmedetomidine group compared with other treatments based on median uptake weights. This may explain why the difference in thalamic activity between connectedness and disconnectedness was smaller with dexmedetomidine than with propofol or sevoflurane.

Exposure to S-ketamine was not associated with similar state-related metabolic effects as with other agents. The dissociative state characteristic of (S-)ketamine is likely caused by preferential antagonism of NMDA receptors in GABAergic interneurons leading to disinhibition of excitatory pyramidal neurons (Homayoun and Moghaddam, 2007; Vollenweider and Kometer, 2010). Consistently, we found a weak state-related effect of increased metabolic activity in left intraparietal sulcus, which is in agreement with previously reported metabolic effects of anesthetic concentrations of S-ketamine (Långsjö et al., 2005). The high metabolism and complexity with ketamine and low metabolism and complexity with other anesthetics has often been associated with their respective high and low probabilities for reporting disconnected conscious experiences on awakening (Lee et al., 2022). This has likely contributed to the common assumption that high metabolism and complexity are necessary preconditions for consciousness. In the current study, disconnected consciousness was frequently reported after low metabolism brain states, challenging the notion that the difference in the observed brain states between ketamine and other anesthetics could be simply related to high or low awareness. While the interpretations related to the neural correlates of consciousness per se are out of the scope of the current study, our findings support the view that disconnectedness can be caused by low and high metabolism mechanisms.

A notable limitation of the current study is that the interviews used for verifying disconnectedness were only obtained at the end of the experiment. Report analysis suggested profound amnesia since awareness of external events was only reported by one responsive subject from the S-ketamine group (Radek et al., 2021) despite frequent signs of connected consciousness implicated by responses to auditory ques. However, based on our previous experiments using serial awakenings with immediate interviews, explicit awareness is rare in unresponsive subjects although internal subjective experiences may still occur rather frequently (Noreika et al., 2011; Radek et al., 2018; Valli et al., 2023). The use of a face mask for sevoflurane administration may have caused relative stimulation in comparison to intravenous anesthetics. While this could have influenced betweengroup comparisons, it is unlikely that the main within-group results were affected. Furthermore, the proportion of disconnected subjects was higher in the sevoflurane group compared with other treatments. Generally, there were more unresponsive than responsive trials, suggesting that we slightly overestimated the EC₅₀ doses. This was especially evident in the sevoflurane and dexmedetomidine groups.

The frequently repeated testing for responsiveness revealed important information about the labile nature of the state of consciousness at the estimated EC_{50} anesthetic exposure. While the state fluctuation within subjects could have reduced the sensitivity to detect state-related differences between subjects, it allowed us to test for associations between the degree of disconnectedness and brain activity. The convergence of the results between the two analysis methods suggests that the effect of the within-subject state fluctuation on the main finding was relatively small.

The most notable strength of the study is the use of four anesthetics with different primary mechanisms of action in the same experimental settings while allowing distinction between the state-related and drug-related effects. In summary, we found consistent evidence that decreased thalamic activity is a specific correlate of disconnectedness for three commonly used anesthetics representing three different primary mechanisms of action. Our results highlight the need for carefully designed studies to elucidate the neural mechanisms associated with specific states of consciousness.

References

- Akeju O, Loggia ML, Catana C, Pavone KJ, Vazquez R, Rhee J, Contreras Ramirez V, Chonde DB, Izquierdo-Garcia D, Arabasz G, Hsu S, Habeeb K, Hooker JM, Napadow V, Brown EN, Purdon PL (2014) Disruption of thalamic functional connectivity is a neural correlate of dexmedetomidine-induced unconsciousness. Elife 3:e04499.
- Alkire MT, Miller J (2005) General anesthesia and the neural correlates of consciousness. Prog Brain Res 150:229–244.
- Alkire MT, Haier RJ, Shah NK, Anderson CT (1997) Positron emission tomography study of regional cerebral metabolism in humans during isoflurane anesthesia. Anesthesiology 86:549–557.
- Alkire MT, Pomfrett CJ, Haier RJ, Gianzero MV, Chan CM, Jacobsen BP, Fallon JH (1999) Functional brain imaging during anesthesia in humans: effects of halothane on global and regional cerebral glucose metabolism. Anesthesiology 90:701–709.
- Alkire MT, Haier RJ, Fallon JH (2000) Toward a unified theory of narcosis: brain imaging evidence for a thalamocortical switch as the neurophysiologic basis of anesthetic-induced unconsciousness. Conscious Cogn 9:370–386.
- Alkire MT, McReynolds JR, Hahn EL, Trivedi AN (2007) Thalamic microinjection of nicotine reverses sevoflurane-induced loss of righting reflex in the rat. Anesthesiology 107:264–272.
- Alkire MT, Hudetz AG, Tononi G (2008) Consciousness and anesthesia. Science 322:876–880.
- Alkire MT, Asher CD, Franciscus AM, Hahn EL (2009) Thalamic microinfusion of antibody to a voltage-gated potassium channel restores consciousness during anesthesia. Anesthesiology 110:766–773.
- Baker R, Gent TC, Yang Q, Parker S, Vyssotski AL, Wisden W, Brickley SG, Franks NP (2014) Altered activity in the central medial thalamus precedes changes in the neocortex during transitions into both sleep and propofol anesthesia. J Neurosci 34:13326–13335.
- Bastos AM, Donoghue JA, Brincat SL, Mahnke M, Yanar J, Correa J, Waite AS, Lundqvist M, Roy J, Brown EN, Miller EK (2021) Neural effects of propofol-induced unconsciousness and its reversal using thalamic stimulation. Elife 10:e60824.
- Bonhomme V, Staquet C, Montupil J, Defresne A, Kirsch M, Martial C, Vanhaudenhuyse A, Chatelle C, Larroque SK, Raimondo F, Demertzi A, Bodart O, Laureys S, Gosseries O (2019) General anesthesia: a probe to explore consciousness. Front Syst Neurosci 13:36.
- Boveroux P, Vanhaudenhuyse A, Bruno M-A, Noirhomme Q, Lauwick S, Luxen A, Degueldre C, Plenevaux A, Schnakers C, Phillips C, Brichant J-F, Bonhomme V, Maquet P, Greicius MD, Laureys S, Boly M (2010) Breakdown of within- and between-network resting state functional magnetic resonance imaging connectivity during propofol-induced loss of consciousness. Anesthesiology 113:1038–1053.
- Brett M, Anton JL, Valabregue R, Poline JB (2002) Region of interest analysis using an SPM toolbox. Presented at the 8th International Conference on Functional Mapping of the Human Brain, June 2–6, 2002, Sendai, Japan.
- Casey CP, Tanabe S, Farahbakhsh Z, Parker M, Bo A, White M, Ballweg T, Mcintosh A, Filbey W, Saalmann Y, Pearce RA, Sanders RD (2022) Distinct EEG signatures differentiate unconsciousness and disconnection during anaesthesia and sleep. Br J Anaesth 128:1006–1018.
- de Jong HWAM, van Velden FHP, Kloet RW, Buijs FL, Boellaard R, Lammertsma AA (2007) Performance evaluation of the ECAT HRRT: an LSO-LYSO double layer high resolution, high sensitivity scanner. Phys Med Biol 52:1505–1526.
- Domino EF, Domino SE, Smith RE, Domino LE, Goulet JR, Domino KE, Zsigmond EK (1984) Ketamine kinetics in unmedicated and diazepampremedicated subjects. Clin Pharmacol Ther 36:645–653.
- Duelli R, Kuschinsky W (2001) Brain glucose transporters: relationship to local energy demand. News Physiol Sci 16:71–76.
- Eriksson L, Wienhard K, Eriksson M, Casey ME, Knoess C, Bruckbauer T, Hamill J, Schmand M, Gremillion T, Lenox M, Conti M, Bendriem B, Heiss WD, Nutt R (2002) The ECAT HRRT: NEMA NEC evaluation of the HRRT system, the new high-resolution research tomograph. IEEE Trans Nucl Sci 49:2085–2088.
- Fan L, Li H, Zhuo J, Zhang Y, Wang J, Chen L, Yang Z, Chu C, Xie S, Laird AR, Fox PT, Eickhoff SB, Yu C, Jiang T (2016) The human brainnetome atlas: a new brain atlas based on connectional architecture. Cereb Cortex 26:3508–3526.
- Fiset P, Paus T, Daloze T, Plourde G, Meuret P, Bonhomme V, Hajj-Ali N, Backman SB, Evans AC (1999) Brain mechanisms of propofol-induced

loss of consciousness in humans: a positron emission tomographic study. J Neurosci 19:5506–5513.

- Friston KJ, Penny WD, Glaser DE (2005) Conjunction revisited. Neuroimage 25:661–667.
- Gallagher BM, Fowler JS, Gutterson NI, MacGregor RR, Wan CN, Wolf AP (1978) Metabolic trapping as a principle of oradiopharmaceutical design: some factors resposible for the biodistribution of [18F] 2-deoxy-2-fluoro-D-glucose. J Nucl Med 19:1154–1161.
- Hill S, Tononi G (2005) Modeling sleep and wakefulness in the thalamocortical system. J Neurophysiol 93:1671–1698.
- Homayoun H, Moghaddam B (2007) NMDA receptor hypofunction produces opposite effects on prefrontal cortex interneurons and pyramidal neurons. J Neurosci 27:11496–11500.
- Johansson J, Keller S, Tuisku J, Teräs M (2016) Performance of an imagebased motion compensation algorithm for the hrrt: a striatum-phantom study with true motion. 2016 IEEE Nuclear Science Symposium, Medical Imaging Conference and Room-Temperature Semiconductor Detector Workshop (NSS/MIC/RTSD), pp 1–3. 29 October - 06 November 2016, Strasbourg, France.
- John ER, Prichep LS (2005) The anesthetic cascade: a theory of how anesthesia suppresses consciousness. Anesthesiology 102:447–471.
- Kaisti KK, Metsähonkala L, Teräs M, Oikonen V, Aalto S, Jääskeläinen S, Hinkka S, Scheinin H (2002) Effects of surgical levels of propofol and sevoflurane anesthesia on cerebral blood flow in healthy subjects studied with positron emission tomography. Anesthesiology 96:1358–1370.
- Kallionpää RE, Valli K, Scheinin A, Långsjö J, Maksimow A, Vahlberg T, Revonsuo A, Scheinin H, Mashour GA, Li D (2020) Alpha band frontal connectivity is a state-specific electroencephalographic correlate of unresponsiveness during exposure to dexmedetomidine and propofol. Br J Anaesth 125:518–528.
- Kaskinoro K, Maksimow A, Långsjö J, Aantaa R, Jääskeläinen S, Kaisti K, Särkelä M, Scheinin H (2011) Wide inter-individual variability of bispectral index and spectral entropy at loss of consciousness during increasing concentrations of dexmedetomidine, propofol, and sevoflurane. Br J Anaesth 107:573–580.
- Laaksonen L, Kallioinen M, Långsjö J, Laitio T, Scheinin A, Scheinin J, Kaisti K, Maksimow A, Kallionpää RE, Rajala V, Johansson J, Kantonen O, Nyman M, Sirén S, Valli K, Revonsuo A, Solin O, Vahlberg T, Alkire M, Scheinin H (2018) Comparative effects of dexmedetomidine, propofol, sevoflurane, and S-ketamine on regional cerebral glucose metabolism in humans: a positron emission tomography study. Br J Anaesth 121:281–290.
- Laitio RM, Långsjö JW, Aalto S, Kaisti KK, Salmi E, Maksimow A, Aantaa R, Oikonen V, Viljanen T, Parkkola R, Scheinin H (2009) The effects of xenon anesthesia on the relationship between cerebral glucose metabolism and blood flow in healthy subjects: a positron emission tomography study. Anesth Analg 108:593–600.
- Lee M, Sanz LRD, Barra A, Wolff A, Nieminen JO, Boly M, Rosanova M, Casarotto S, Bodart O, Annen J, Thibaut A, Panda R, Bonhomme V, Massimini M, Tononi G, Laureys S, Gosseries O, Lee SW (2022) Quantifying arousal and awareness in altered states of consciousness using interpretable deep learning. Nat Commun 13:1064.
- Långsjö JW, Maksimow A, Salmi E, Kaisti K, Aalto S, Oikonen V, Hinkka S, Aantaa R, Sipilä H, Viljanen T, Parkkola R, Scheinin H (2005) S-ketamine anesthesia increases cerebral blood flow in excess of the metabolic needs in humans. Anesthesiology 103:258–268.
- Långsjö JW, Alkire MT, Kaskinoro K, Hayama H, Maksimow A, Kaisti KK, Aalto S, Aantaa R, Jääskeläinen SK, Revonsuo A, Scheinin H (2012) Returning from oblivion: imaging the neural core of consciousness. J Neurosci 32:4935–4943.
- Långsjö JW, Lehtimäki K, Ruohonen J, Sajanti A, Sandell S, Heikkilä HT, Brander A, Saarinen K, Herrala L (2016) Critical neural targets for (the level of) human consciousness: arousal arrest and unconsciousness after sumatriptan administration. Brain Inj 30:1731–1736.
- Lichtner G, Auksztulewicz R, Kirilina E, Velten H, Mavrodis D, Scheel M, Blankenburg F, von Dincklage F (2018) Effects of propofol anesthesia on the processing of noxious stimuli in the spinal cord and the brain. Neuroimage 172:642–653.
- Liu X, Lauer KK, Ward BD, Li S-J, Hudetz AG (2013) Differential effects of deep sedation with propofol on the specific and nonspecific thalamocortical systems: a functional magnetic resonance imaging study. Anesthesiology 118:59–69.

- Kantonen et al. Neural Correlates of Disconnectedness
- Llinás RR, Steriade M (2006) Bursting of thalamic neurons and states of vigilance. J Neurophysiol 95:3297–3308.
- Mai J, Majtanic M, Paxinos G (2016) Atlas of the human brain, Ed 4. London: Academic.
- Marsh B, White M, Morton N, Kenny GN (1991) Pharmacokinetic model driven infusion of propofol in children. Br J Anaesth 67:41–48.
- Mashour GA, Alkire MT (2013) Consciousness, anesthesia, and the thalamocortical system. Anesthesiology 118:13–15.
- McKinstry-Wu AR, Wasilczuk AZ, Harrison BA, Bedell VM, Sridharan MJ, Breig JJ, Pack M, Kelz MB, Proekt A (2019) Analysis of stochastic fluctuations in responsiveness is a critical step toward personalized anesthesia. Elife 8:e50143.
- Noreika V, Jylhänkangas L, Móró L, Valli K, Kaskinoro K, Aantaa R, Scheinin H, Revonsuo A (2011) Consciousness lost and found: subjective experiences in an unresponsive state. Brain Cogn 77:327–334.
- Nourski KV, Steinschneider M, Rhone AE, Kawasaki H, Howard MA 3rd, Banks MI (2018) Auditory predictive coding across awareness states under anesthesia: an intracranial electrophysiology study. J Neurosci 38:8441–8452.
- Pal D, Li D, Dean JG, Brito MA, Liu T, Fryzel AM, Hudetz AG, Mashour GA (2020) Level of consciousness is dissociable from electroencephalographic measures of cortical connectivity, slow oscillations, and complexity. J Neurosci 40:605–618.
- Palanca BJA, Mitra A, Larson-Prior L, Snyder AZ, Avidan MS, Raichle ME (2015) Resting-state functional magnetic resonance imaging correlates of sevoflurane-induced unconsciousness. Anesthesiology 123:346-356.
- Phelps ME, Huang SC, Hoffman EJ, Selin C, Sokoloff L, Kuhl DE (1979) Tomographic measurement of local cerebral glucose metabolic rate in humans with (F-18)2-fluoro-2-deoxy-D-glucose: validation of method. Ann Neurol 6:371–388.
- Radek L, Kallionpää RE, Karvonen M, Scheinin A, Maksimow A, Långsjö J, Kaisti K, Vahlberg T, Revonsuo A, Scheinin H, Valli K (2018) Dreaming and awareness during dexmedetomidine- and propofol-induced unresponsiveness. Br J Anaesth 121:260–269.
- Radek L, Koskinen L, Sandman N, Laaksonen L, Kallionpää R, Scheinin A, Rajala V, Maksimow A, Laitio T, Revonsuo A, Scheinin H, Valli K (2021) On no man's land: subjective experiences during unresponsive and responsivesedative states induced by four different anesthetic agents. Conscious Cogn 96:103239.
- Redinbaugh MJ, Phillips JM, Kambi NA, Mohanta S, Andryk S, Dooley GL, Afrasiabi M, Raz A, Saalmann YB (2020) Thalamus modulates consciousness via layer-specific control of cortex. Neuron 106:66–75.e12.
- Saalmann YB (2014) Intralaminar and medial thalamic influence on cortical synchrony, information transmission and cognition. Front Syst Neurosci 8:83.
- Scheinin A, Kallionpää RE, Li D, Kallioinen M, Kaisti K, Långsjö J, Maksimow A, Vahlberg T, Valli K, Mashour GA, Revonsuo A, Scheinin H (2018a) Differentiating drug-related and state-related effects of dexmedetomidine and propofol on the electroencephalogram. Anesthesiology 129:22–36.
- Scheinin H, Alkire EC, Scheinin A, Alkire MT, Kantonen O, Långsjö J (2018b) Using positron emission tomography in revealing the mystery of general anesthesia: study design challenges and opportunities. Methods Enzymol 603:279–303.
- Scheinin A, Kantonen O, Alkire M, Långsjö J, Kallionpää RE, Kaisti K, Radek L, Johansson J, Sandman N, Nyman M, Scheinin M, Vahlberg T, Revonsuo A, Valli K, Scheinin H (2021) Foundations of human consciousness: imaging the twilight zone. J Neurosci 41:1769–1778.
- Schmitt LI, Wimmer RD, Nakajima M, Happ M, Mofakham S, Halassa MM (2017) Thalamic amplification of cortical connectivity sustains attentional control. Nature 545:219–223.
- Talke P, Lobo E, Brown R (2003) Systemically administered alpha2agonist-induced peripheral vasoconstriction in humans. Anesthesiology 99:65–70.
- Tasserie J, Uhrig L, Sitt JD, Manasova D, Dupont M, Dehaene S, Jarraya B (2022) Deep brain stimulation of the thalamus restores signatures of consciousness in a nonhuman primate model. Sci Adv 8:eabl5547.
- Valli K, Radek L, Kallionpää RE, Scheinin A, Långsjö J, Kaisti K, Kantonen O, Korhonen J, Vahlberg T, Revonsuo A, Scheinin H (2023) Subjective experiences during dexmedetomidine- and propofol-induced unresponsiveness and non-rapid eye movement sleep in healthy male subjects. Br J

Anaesth. Advance online publication. Retrieved May 31, 2023. https://doi.org/10.1016/j.bja.2023.04.026.

- Varrone A, Asenbaum S, Vander Borght T, Booij J, Nobili F, Någren K, Darcourt J, Kapucu OL, Tatsch K, Bartenstein P, Van Laere K; European Association of Nuclear Medicine Neuroimaging Committee (2009) EANM procedure guidelines for PET brain imaging using [18F]FDG, version 2. Eur J Nucl Med Mol Imaging 36:2103–2110.
- Vijayan S, Ching S, Purdon PL, Brown EN, Kopell NJ (2013) Thalamocortical mechanisms for the anteriorization of α rhythms during propofolinduced unconsciousness. J Neurosci 33:11070–11075.
- Vollenweider FX, Kometer M (2010) The neurobiology of psychedelic drugs: implications for the treatment of mood disorders. Nat Rev Neurosci 11:642–651.
- Ward LM (2011) The thalamic dynamic core theory of conscious experience. Conscious Cogn 20:464–486.
- Warnaby CE, Seretny M, Ní Mhuircheartaigh R, Rogers R, Jbabdi S, Sleigh J, Tracey I (2016) Anesthesia-induced suppression of human dorsal anterior insula responsivity at loss of volitional behavioral response. Anesthesiology 124:766–778.

- Wasilczuk AZ, Harrison BA, Kwasniewska P, Ku B, Kelz MB, McKinstry-Wu AR, Proekt A (2020) Resistance to state transitions in responsiveness is differentially modulated by different volatile anaesthetics in male mice. Br J Anaesth 125:308–320.
- White NS, Alkire MT (2003) Impaired thalamocortical connectivity in humans during general-anesthetic-induced unconsciousness. Neuroimage 19:402– 411.
- Wu HM, Bergsneider M, Glenn TC, Yeh E, Hovda DA, Phelps ME, Huang SC (2003) Measurement of the global lumped constant for 2-deoxy-2-[18F]fluoro-D-glucose in normal human brain using [15O]water and 2-deoxy-2-[18F]fluoro-D-glucose positron emission tomography imaging. A method with validation based on multiple methodologies. Mol Imaging Biol 5:32–41.
- Xie G, Deschamps A, Backman SB, Fiset P, Chartrand D, Dagher A, Plourde G (2011) Critical involvement of the thalamus and precuneus during restoration of consciousness with physostigmine in humans during propofol anaesthesia: a positron emission tomography study. Br J Anaesth 106:548–557.